Author's response to reviews

Title: Investigation of an Influenza A (H3N2) Outbreak in Evacuation Centres following the Great East Japan Earthquake, 2011

Authors:

Taro Kamigaki (kamigakit@med.tohoku.ac.jp)
Jin Seino (seino916@mnh.go.jp)
Kentaro Tohma (tohma-org@med.tohoku.ac.jp)
Nao Nukiwa-Soma (naons@med.tohoku.ac.jp)
Kanako Otani (otanik@med.tohoku.ac.jp)
Hitoshi Oshitani (oshitanih@med.tohoku.ac.jp)

Version: 3 Date: 11 October 2013

Author's response to reviews: see over
11 October, 2013

Dr Natalie Pafitis  
Exececutive Editor, BMC Public Health  
BioMed Central  
236 Gray's Inn Road, London WC1X 8HB

Dear Dr Pafitis

Please find enclosed our revised manuscript titled “Investigation of an Influenza A (H3N2) Outbreak in Evacuation Centres following the Great East Japan Earthquake, 2011” for consideration as a Research Articles for publication in *BMC Public Health*.

With the advice from the editorial office of *BMC Public Health* to confirm the ethical status for our study, we submitted our study protocol to the ethical committee for the evaluation and obtained the approval. In accordance, we revised our manuscript by including the section of ethical consideration.

We appreciate your review of this revised manuscript for publication in *BMC Public Health*.

Sincerely,

Taro Kamigaki, M.D.

Department of Virology  
Tohoku University Graduate School of Medicine  
2-1 Seiryo machi, Aoba-ku  
Sendai, 9808575, Japan  
(Tel) +81-22-7178213  
(FAX) +81-22-7178212  
(Email) kamigakit@med.tohoku.ac.jp